Author’s response to reviews

Title: Operationalization of bi-directional screening for tuberculosis and diabetes in private sector healthcare clinics in Karachi, Pakistan

Authors:

Mashal Basir (mashal.basir@gmail.com;mashal.basir@columbia.edu)
Shifa Habib (shifa.habib@chshealthcare.org)
Syed Zaidi (asad.zaidi@chshealthcare.org)
Saira Khowaja (saira.khowaja@ird.global)
Hamidah Hussain (hamidah.hussain@irdresearch.org)
Rashida Ferrand (rashida.ferrand@lshtm.ac.uk)
Aamir Khan (aamir.khan@irdresearch.org)

Version: 4 Date: 13 Feb 2019

Author’s response to reviews:

Editor Comments:

1. In the Ethics approval and consent to participate section, please confirm that your ethics committee formally approved using verbal consent and how the verbal consent was documented.

2. Please be sure to combine the statements included in the "Statement of Ethics" and "Ethics, Consent and Permissions" sections into one section with the heading "Ethics approval and consent to participate section."

Author response:

I have confirmed that our IRB committee formally approved verbal consent and have noted down where it was recorded. I have also combined the statements into an "ethics approval and consent to participate" section on page 17-18 lines 362-367.